Author's response to reviews

Title: Predictors of labor abnormalities in university hospital: unmatched case control study

Authors:

Wayou Abraham (wayouabrham@gmail.com)
Yifru Berhan (yifrub@yahoo.com)

Version: 5 Date: 27 November 2013

Author's response to reviews:

Section editor comment

“Please confirm that the adjusted odds ratio its confidence interval for "Pelvic status: unknown" is 0.5(0.31-0.79) indeed, and not 0.5(0.31-1.79). It is curious that the CI for the adjusted OR should be that different from the CI for the crude odds ratio, which is reported as 0.5(0.30-1.73). Do we have a typo here?”

Response: Actually, the analyses were done by both investigators independently before write up and submission of this manuscript. What I mean; we have checked the consistency of our analysis findings; it was not typing error.

Yifru Berhan